Author’s response to reviews

Title: Shear stress improves the Endothelial Progenitor Cell function via the CXCR7/ERK pathway axis in the Coronary Artery Disease cases

Authors:

Hua Zhou (zhouhua1022@sina.com)
Qiang Tu (609538829@qq.com)
Yan Zhang (11259662@qq.com)
Hua Qiang Xie (xiehqiang@126.com)
Qing Yun Shuai (280540383@qq.com)
Xiao Chuan Huang (1633805294@qq.com)
Jie Fu (m15897856680@163.com)
Zheng Cao (caozheng908@163.com)

Version: 2 Date: 15 Mar 2020

Author’s response to reviews:

We have submitted the article and pictures as you requested. Thanks.